Author's response to reviews

Title: Injuries associated with housing conditions in Europe: a burden of disease study based on 2004 injury data

Authors:

Michael D Keall (michael.keall@otago.ac.nz)
David Ormandy (David.Ormandy@warwick.ac.uk)
Michael G Baker (michael.baker@otago.ac.nz)

Version: 3 Date: 14 October 2011

Author's response to reviews:

Response to reviews of “Estimating the burden of home injury due to housing conditions in Europe”

Thanks to the editor and to the reviewers

Editorial comments:
As for the formatting, it does look better. Before resubmitting, please note that all the lines in the References section should be aligned i.e. the subsequent lines under the first should not be indented.

We have changed the formatting accordingly

Reviewer's report
Title: Injuries associated with housing conditions in Europe: a burden of disease study
Version: 2 Date: 12 August 2011
Reviewer: bridget kool

Reviewer's report:
Dear authors, the revisions made to the paper now give the manuscript a better flow and showcase the findings more succinctly. I have no further comments.

Level of interest: An article of importance in its field
Quality of written English: Acceptable
Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests:
Nil competing interests

Reviewer's report
Title: Injuries associated with housing conditions in Europe: a burden of disease study
Version: 2 Date: 11 October 2011
Reviewer: Hossein kazemeini
Reviewer’s report:
I think they have modified it according the comments. The only point is that they haven not consider the year of study in the title.
We have now changed the title to include year